
JUC, 22-04-2016, 13.30, Arpad's room

Des: everything works
Harro: still working on weight plots, fringe plots work but still need to be
started up automatically, will look into that

Arpad will talk to dr. Bob and operators, start set of correlations of NMEs
next week. Des will have to demonstrate how to start up stuff, have
operators do the rest. Maybe run the pipeline as well at some point. Lack
of scientific support is an issue.


